Author’s response to reviews

Title: Resting heart rate and impaired glucose regulation in Chinese middle-aged and elderly people: a cross-sectional analysis

Authors:

Zhen Yang (yangzhen1020@hotmail.com)
Weiwei Zhang (amanda_zww@126.com)
Xiaoyong Li (lixiaoyong@126.com)
Lingfei Zhu (zhuzhu@163.com)
Yixin Niu (niuniu@126.com)
Ning Lin (linning@163.com)
Shuai Lu (lushuai@163.com)
Hongmei Zhang (zanghongmei@126.com)
Xuanchun Wang (wangxunchun@126.com)
Jie Wen (wenjie@163.com)
Guang Ning (ningguang@126.com)
Li Qin (qinli@126.com)
Qing Su (suqingxinhua@163.com)

Version: 2 Date: 14 Jul 2017

Author’s response to reviews:

Dear Editor:

Our manuscript has been extensively revised in accordance with the enclosed reviewer’s comments. All changes in manuscripts emphasized by red characters.
Reviewer reports:

Stefan Söderberg (Reviewer 1): The manuscript "Resting Heart …" by Yang and colleagueus has improved and the authors have been sensitive to our comments. However, I still have some points to be addressed.

1. Still some passages indicating causality, like line 27 and line 42, both on page saying, "significant contributor". This is not correct.

   Answers:

   We thanks for reviewer’s reminder. According to reviewer’s instruction, we have revised the expression in the manuscript.

2. Line 56, page 11, "first study", which obviously is not true. Please rephrase.

   Answers:

   We thanks for reviewer’s reminder. According to reviewer’s instruction, we have revised the expression in the manuscript.

3. I strongly suggest that the authors should consult a professional language bureau for appropriate language revision, there are still too many grammar mistakes.

   Answers:

   We thanks for reviewer’s reminder. According to reviewer’s instruction, we have consulted professional language bureau for appropriate language revision.